Dear Reviewer

We greatly appreciate your professional commentary work on our articles. As you are concerned, there are several issues that need to be addressed. Based on your good suggestion, we have made extensive corrections to the previous draft, the corrections have been marked in red, and the specific corrections are as follows:

Reviewer 1

1. Added the relevant content of situs inversus totalis in the Introduction, and added a description of whether the infant's death was related to situs inversus totalis.

2. Added author contributions.

The new content has been edited again in non-native languages.

Situs inversus totalis is a rare congenital malformation, and some patients may also suffer from defective ciliary motility. The ciliary immobility is involved in the absence of mucociliary transport in the respiratory epithelia, which may induce lung infections. We here report a patient who suffered from rare congenital tuberculosis, tuberculous meningitis, and total visceral reversal. Congenital tuberculosis complicated with situs inversus totalis was not found from the Google Scholar and PubMed databases.

Reviewer 2

Thanks for your comments.

KS is also known as ciliary immobility syndrome, which can lead to obstruction of mucus drainage from the respiratory tract, which increases the possibility of lung infection. However, we cannot confirm whether the baby has KS and whether KS will increase the prevalence of congenital tuberculosis.